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14 June 2009
MEMORANDUM FOR CENWP-OP, ATTN: Operations Biologist
SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for Week 24 of 2009, which covers the period from 07 through 13 June 2009.
1 OPERATION SUMMARY

1.1 Daily average river flows ranged from 257.6 to 360.6 kcfs.  Daily average project tailwater ranged from 20.4’ to 26.0’ msl.  Daily average powerhouse forebay elevation ranged from 73.8’ to 75.5’ msl.  Daily average river temperature ranged from 61ºF to 62ºF.  Secchi disk measurements ranged from 3.0’ to 4.5’.
1.2 Daily average spill ranged from 99.7 to 164.2 kcfs.
1.2.1 Spillbay 17 remains on two dogs (2.9’).  The bay should be at 3.5’ but due to a broken hoist it has to be dogged instead.  Flow is compensated through adjacent bays.
1.3 Unit Operation.  Powerhouse 2 remains the priority powerhouse.
1.3.1 Unit 7 has remained out of service since 04 June 2007 for turbine rehab.

1.3.2 Unit 9 has remained out of service since 04 August 2008 for turbine rehab.
1.3.3 Unit 11 has remained out of service since 12 June 2008 due to rotor damage.
1.3.4 Unit 16 has remained out of service since 09 May due to exciter troubles and a grounding issue.

1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore (WS) fish ladder.  The units may be placed on standby, as necessary, during nighttime hours to float debris.
1.5 Gatewell drawdowns are measured on Wednesdays.
2 ACTIVITIES

2.1 Auxiliary Water System Closures.  The following AWS valves were closed this reporting week:

2.1.1 FV 3-7 was closed for cleaning from 1311 until 1456 on 11 June.
2.1.2 FV 3-9 was closed for cleaning from 1311 until 1456 on 11 June.

2.1.3 FV 5-9 was closed for cleaning from 0730 until 0745 on 11 June.

2.1.4 FV 6-9 was closed for cleaning from 1313 until 1325 on 11 June.

2.2 STS/VBS inspections.  The following STS/VBS inspections occurred this reporting week:  
Table 1.  STS and VBS inspections.  
	Unit
	Date
	Run Hours
	Comments

	1-2
	NA
	NA
	Remain out of service until 3 September.

	11
	NA
	NA
	Currently out of service

	12
	
	
	 VBSs cleaned on 08, 11, 13 June

	13
	
	
	VBSs cleaned on 08, 11, 13 June

	14 
	
	
	 VBSs cleaned on 08, 10, 11, 13 June

	15
	
	
	VBSs cleaned on 08, 10, 11, 13 June

	16
	
	
	Currently out of service

	17
	
	
	VBSs cleaned on 08, 10, 11, 13 June

	18
	
	
	VBSs cleaned on 08, 10, 11, 13 June


2.2.1
VBSs were cleaned over the weekend to keep drawdown within FPP criteria.

2.3 Dewatering and Fish Salvage.  No fish salvage or dewatering activities occurred this week.
3 RESEARCH  

3.1 CRITFC - Annual chinook, coho, sockeye and steelhead sampling in the AFF continues.

3.2 CRITFC- JMF video.  Waiting on new equipment for monitoring juvenile fish passage.  
3.3 CTUIR.  Collected 14 lamprey for radio-tagging from the AFF on 07 June.  

3.4 NOAA-FGE.  Test releases of PIT-tagged run of the river fish continued this week to assess passage conditions at two turbine operating ranges (middle and upper end of the 1% efficiency range).  No test fish were released this week, testing will resume with sub-yearling the week of 14 June.
3.5 NOAA Lamprey.  Trapping lamprey at the AFF for passage evaluations.    
3.6 NOAA- pathogens. Determination of pathogen prevalence in juvenile salmonids.  Subyearling collection to begin the week of 14 June.
3.7 NOAA- PIT tag detectors.  Vibration sensors remain installed on Spill bay 6. 

3.8 PNNL- BGS.  All hydrophones are installed and collecting data for the BGS evaluation.  
3.9 PSMFC- Smolt monitoring.  Daily condition and index monitoring continues.  Fallbacks recorded crossing the separator bars included; three steelhead two salmon, and one unknown.  Gas Bubble Trauma (GBT) exams occur two days per week at the JMF during spill; see www.fpc.org/CurrentDaily/Gbtsum.txt for further details.  No fish collected for other researchers this week.
3.10 WDFW Fish Counting.  Visual counting continues at Bradford Island and Washington Shore fishways.
3.11 WDFW/ODFW/CRITFC Sea lion hazing.  Hazing from boats and trapping concluded. 
3.12 USDA Wildlife Services. Hazing sea lions from shore concluded.  Avian hazing continues.

3.13 University of Idaho.  Tagging adult spring Chinook for evaluation of the Cascades Is. entrance weir.
3.14 University of Washington.  Barge versus bypass fish testing for latent mortality continues.  Barge and bypass collection has concluded for the season.
4 FISHWAYS 

4.1 Project biologists inspected 07 through 13 June.  NOAA Fisheries inspected on 12 June.
4.2  Fish Passage Plan Observations.
Table 2.  FPP Out of Criteria items.
	DATE
	LOCATion
	FPP itEMS
	CAUSE
	RESPONSE
	TR#

	BRADFORD ISLAND (BI) 

	6/9-11
	FV3-9/FV3-7
	Trashrack diff.
	Debris
	Riggers cleaned
	

	CASCADES ISLAND (CI) – nothing out of criteria

	
	
	
	
	
	

	WASHINGTON SHORE (WS) – nothing out of criteria

	
	
	
	
	
	

	ITS – nothing out of criteria

	DSM2/ JMF- nothing out of criteria


4.2.1 PH1 PCC diffusers FG2-22A and FG2-19 are closed or partially open.  Repairs scheduled for 09/10.  

4.2.2 Design of a mechanical velocity meter at the PH2 south monolith is in progress.  

4.2.3 CI south entrance gate remains broken and closed.  It is been OOS since 10 April.  It will remain in the closed position for all tailwater elevations during the modified entrance evaluation. 
4.2.4 The NUE staff gauge is missing above 24’ msl.  Water elevation is estimated.

4.3 Fishway Temperatures.  Measurements are taken three weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington Shore.  

4.3.1 The WS and BI probes need to be calibrated.  The AFF probe is being used as a surrogate.
Table 3.  Bonneville Fishway Temperatures

	Date
	AFF average

	6/07
	59.3

	6/08
	59.5

	6/09
	60.0

	6/10
	60.3

	6/11
	60.4

	6/12
	60.3

	6/13
	60.7


4.4 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week.  
4.5 Adult Fishways. 
4.5.1 Calibration was checked on 13 June.  
4.5.2 The Washington Shore fishway entrances are in manual due to SA-24 board calibration issues.  The operators are adjusting the gates to maintain at least a 13’ opening.

4.5.3 There are currently 11 of 12 floating orifice gates (FOG) installed at Powerhouse 2.  The last gate was removed for maintenance and replaced with a bulkhead.
4.5.4 All adult fishways remain in shad mode.
4.5.5 SLEDs remain out.
4.6 Juvenile Fishways.
4.6.1 Chain gate 1C remains closed and 1B remains open to the sill (68’ msl).  The automated gates are set at 70’ msl and are not yet automated. Fisheries has observed an increase in debris in the ITS and reduced water velocities, potentially due to the removal of chain gate 1B.  Fisheries will try other gate configurations to increase velocity through the channel, particularly north of the wing wall.
4.7 Lamprey Fishways.  
4.7.1 The CI entrance weir and bulkhead slot filler are installed.  
4.7.2 All lamprey passage structures remain watered-up.   
5 CONSTRUCTION
5.1 PH1 Ice and Trash Sluiceway Wall Removal.  Wall removal is scheduled for FY10.  Old DSM bypass equipment in the tailrace is scheduled for removal in FY10 as well.
5.1.1 The Program Logic Control (PLC) for the automated gates is still being worked on.  All gates continue to operate in manual.
5.2 TIE crane.  The new boom is to arrive in mid-June; load testing should occur by the first week in July.
6 HAZMAT, SPILLS AND CLEANUP.   Nothing to report.
7 TELETYPES AND OTHER NOTEWORTHY ITEMS
7.1 A teletype sent on 09 April requests spring spill to begin at 0001 on 10 April, continuing until further notice.

7.2 A teletype sent on 17 May and revised on 18 May provides guidance for voluntary and involuntary spill by project as well as spill caps per project for various total dissolved gas (TDG) levels.

7.3 A teletype sent on 12 June requests that from 0600 on 16 June until 2400 on 18 June, the forebay elevation be maintained at 75.0’ – 76.5’ msl. as a hard constraint and 75.5’ – 76.5’ msl. as a soft constraint.  It also requests that from 0001 on 19 June until 1800 on 19 June, the forebay elevation be maintained at 74.5’ – 76.0’ msl. as a hard constraint and 75.0’ – 76.0’ msl as a soft constraint.  This operation is requested to reduce debris in tribal fishing nets and reduce entanglement of fishing nets by limiting pool fluctuation.  

JAMES R. MAHAR, P.E.

Operations Manager
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